Author's response to reviews

Title: Preservice Laboratory Education Strengthening Enhances Sustainable Laboratory Workforce in Ethiopia

Authors:

Peter N Fonjungo (pdf4@cdc.gov)
Yenew Kebede (Kebede@et.cdc.gov)
Wendy Arneson (wendy.arneson@ascp.org)
Dereese Tefera (Teferad@et.cdc.gov)
Kedir Yimer (Yimerk@et.cdc.gov)
Samuel Kinde (samuelkinde@yahoo.com)
Meseret Alem (mesey4839@yahoo.com)
Waqtola Cheneke (waqtolahalt@yahoo.com)
Habtamu Mitiku (habtemit@gmail.com)
Endale Tadesse (endaletd@yahoo.com)
Aster Tsegaye (tsegayeaster@yahoo.com)
Thomas Kenyon (tak8@cdc.gov)

Version: 4 Date: 30 August 2013

Author's response to reviews: see over
Re: MS: 1126792169945906 - Preservice Laboratory Education Strengthening
Enhances Sustainable Laboratory Workforce in Ethiopia

August 30, 2013

Corresponding author:

Peter N. Fonjungo, Ph.D.
Division of Global HIV/AIDS Program, Centers for Global Health
Centers for Disease Control and Prevention (CDC)
1600 Clifton Road, MS G-19
Atlanta, GA 30333
Tel: 404 6394038
Email: pfonjungo@cdc.gov

Dear Editors,

On behalf of all the authors I would like to thank the referees for reviewing the revised
manuscript and recommending two discretionary revisions. We have implemented the
discretionary revisions in tracked changes as recommended as well as provided page and line
numbers for easy referencing (See attached uploaded revised version). Please see below
specific responses to the referee’s discretionary comments. We hope the implemented
recommendations are acceptable. Thank you and do not hesitate to contact to me for any
additional clarifications.

Sincerely,

Peter Fonjungo
Referee 1: Helen Nabwera

No comments

Referee 2: Bernadette Rodak

Discretionary revisions

1. We have deleted the word “of” (page 9, line 1 from top).
2. We have rephrased the last sentence under ‘Study design’ in Methods section as recommended (page 9, lines 13-15 from top).